Author's response to reviews

Title: Interventions to prevent disability in frail community-dwelling elderly: a systematic review

Authors:
Ramon R Daniels (r.daniels@hszuyd.nl)
Erik E van Rossum (e.van.rossum@hszuyd.nl)
Luc L de Witte (l.de.witte@hszuyd.nl)
Ruud G.I.J.M Kempen (g.kempen@zw.unimaas.nl)
Wim W van den Heuvel (heuvelwim@hotmail.com)

Version: 2 Date: 4 December 2008

Author's response to reviews: see over
December 4, 2008

Dear Dr. M. Norton,

We hereby submit a revised manuscript entitled “Interventions to prevent disability in frail community-dwelling older persons; a systematic review” for publication in BMC Health Services Research.

From the e-mail send to us on the 4th of November we have understood that only two minor editorial request have to be met. Both concerning the title page. We have changed the title page according to the “example title page” as presented on your website.

Thank you to accept our manuscript for publication in BMC Health Services Research.

Yours sincerely,

Ramon Daniels, MSc. OT

Erik van Rossum, PhD

Luc de Witte, MD, PhD

Gertrudis Kempen, PhD

Wim van den Heuvel, PhD